Author's response to reviews

Title: Vocational rehabilitation services for patients with cancer: design of a feasibility study incorporating a pilot randomised controlled trial among women with breast cancer following surgery

Authors:

Richard G Kyle (richard.kyle@stir.ac.uk)
Bill Culbard (bill.culbard@hotmail.co.uk)
Josie Evans (josie.evans@stir.ac.uk)
Nicola Gray (n.gray@abdn.ac.uk)
Gill Hubbard (gill.hubbard@stir.ac.uk)

Version: 2 Date: 27 October 2010

Author's response to reviews:

Dear Sir/Madam,

Vocational rehabilitation services for patients with cancer: design of a feasibility study incorporating a pilot randomised controlled trial among women with breast cancer following surgery

I am submitting a revision to this manuscript in order to upload the additional information requested relating to ethical approval and funding.

I have also uploaded a revised version of the manuscript to add a co-author's middle initial.

Yours sincerely,

Dr Richard G Kyle
Research Fellow
Cancer Care Research Centre
University of Stirling